Author's response to reviews

Title: A Systematic Review and Meta-Analysis: Probiotics in the treatment of Irritable Bowel Syndrome

Authors:

Nourieh Hoveyda (n.hoveyda@doctors.org.uk)
Carl Heneghan (carl.heneghan@dphpc.ox.ac.uk)
Kamal Mahtani (krmahntani@hotmail.com)
Rafael Perera (rafael.perera@dphpc.ox.ac.uk)
Nia roberts (nia.roberts@hcl.ox.ac.uk)
Paul Glasziou (paul.glasziou@dphpc.ox.ac.uk)

Version: 5 Date: 9 December 2008

Author's response to reviews: see over
Dr Scott Edmunds
Senior Editor
BMC-series Journals

MS: 3427674432006792
A Systematic Review and Meta-Analysis: Probiotics in the treatment of Irritable Bowel Syndrome
Nourieh Hoveyda, Carl Heneghan, Kamal Mahtani, Rafael Perera, Nia Roberts and Paul Glasziou

Dear Dr Edmunds

Thank you very much for your email. Attached are our replies to the reviewer’s comments, tables and figures and a revised manuscript.

We hope to hear from you in a near future.

Yours Sincerely

Nourieh Hoveyda